Author’s response to reviews

Title: Early Antenatal Care and Knowledge of Partner HIV Status are Associated with Facility Delivery among HIV-infected Women in Western Kenya; A Community-Based Assessment

Authors:

John Kinuthia (kinuthia@u.washington.edu)
Pamela Kohler (pkohler2@u.washington.edu)
John Okanda (JOkanda@kemricdc.org)
George Olilo (golilo@kemricdc.org)
Frank Odhiambo (fodhiambo@kemricdc.org)
Grace John-Stewart (gjohn@u.washington.edu)

Version: 2 Date: 2 June 2014

Author's response to reviews: see over
2nd June 2014

Dear Editor,

RE: REVISED MANUSCRIPT SUBMITTED FOR REVIEW

We wish to submit the revised version the manuscript entitled “Early Antenatal Care and Knowledge of Partner HIV Status are Associated with Facility Delivery among HIV-infected Women in Western Kenya; A Community-Based Assessment” for review and publication in the journal BMC Pregnancy and Childbirth.

We have included line and page numbers and a section on ‘Competing interests’ as advised.

Sincerely,

Dr John Kinuthia MBChB, MMed, MPH
Consultant Obstetrician & Gynaecologist
Head, Research and Program
Kenyatta National Hospital